Author's response to reviews

Title: Presence of intratumoral platelets is associated with tumor vessel structure and metastasis

Authors:

- Rong Li (jbwucn@163.com)
- Meiping Ren (jbwucn@163.com)
- Ni Chen (jbwucn@163.com)
- Mao Luo (jbwucn@163.com)
- Xin Deng (jbwucn@163.com)
- Jiyi Xia (jbwucn@163.com)
- Guang Yu (jbwucn@163.com)
- Jinbo Liu (jbwucn@163.com)
- Bing He (jbwucn@163.com)
- Xu Zhang (jbwucn@163.com)
- Zhuo Zhang (jbwucn@163.com)
- Xiao Zhang (jbwucn@163.com)
- Bing Ran (jbwucn@163.com)
- Jianbo Wu (wuji@missouri.edu)

Version: 8
Date: 30 January 2014

Author's response to reviews:

Dear Editor:

Please consider our manuscript for publication as a regular article in BMC Cancer. We appreciate the thorough review our initial submission (Ms. Number 1617500580110134) received. We added new data and modified the manuscript in response to the reviewers' excellent suggestions. Accordingly, we have modified the title of the revised manuscript to better reflect its revised content.

All of the authors have contributed significantly to this work and have reviewed and approve of this submission. None of the authors has any conflicts of interest to disclose. We have not submitted this work for publication elsewhere.

Thank you very much for considering our work.

Jianbo Wu, MD, PhD